Author’s response to reviews

Title: What, why and how do health systems learn from one another? Insights from eight low and middle income country case studies

Authors:

Sophie Witter (sophiewitter@blueyonder.co.uk; switter@qmu.ac.uk)
Ian Anderson (ian.anderson.economics@gmail.com)
Peter Annear (pannear@unimelb.edu.au)
Abiodun Awosusi (abiodun.awosusi@gmail.com)
Nitin Bhandari (nnbhandari1@gmail.com)
Nouria Brikci (nouria.brikci@opml.co.uk)
Blandine Binachon (blandinebinachon@hotmail.com)
Tata Chanturidze (Tata.Chanturidze@opml.co.uk)
Katherine Gilbert (katherine.gilbert@unimelb.edu.au)
Charity Jensen (charity.jensen@gmail.com)
Tomas Lievens (tomas.lievens@opml.co.uk)
Barbara McPake (barbara.mcpake@unimelb.edu.au)
Snehashish Raichowdhury (sraichowdhury@gmail.com)
Alex Jones (alex.jones@opml.co.uk)

Version: 1 Date: 19 Dec 2018

Author’s response to reviews:

Many thanks for these very helpful comments. We have responded to them as far as we could, within the constraints of our material.

Please see the attached cover letter for a detailed response.

Many thanks,

Sophie & co-authors